Author's response to reviews

Title: Completion of the modified World Health Organization (WHO) partograph during labour in public health institutions of Addis Ababa, Ethiopia

Authors:

Engida Yisma (engiday@gmail.com)
Berhanu Dessalegn (berhanudessa@yahoo.com)
Ayalew Astatkie (ayalewastatkie@rocketmail.com)
Nebreed Fesseha (nfesseha@jsi-ltenk.org.et)

Version: 3 Date: 10 April 2013

Author's response to reviews: see over
Completion of the modified World Health Organization (WHO) partograph during labour in public health institutions of Addis Ababa, Ethiopia (MS: 3708683529179122).

Authors:

Engida Yisma (engiday@gmail.com)
Berhanu Dessalegn (berhanudessa@yahoo.com)
Ayalew Astatkie (ayalewastatkie@rocketmail.com)
Nebreed Fesseha (nonsenseha@jsi-ltenk.org.et)

To: The Editor-in-Chief of the Reproductive Health Journal,

Dear Jose Belizan,

Thank you so much for your constructive feedback! We have shortened the “Introduction” and “Discussion” sections by merging paragraphs and sentences which have similar concepts. We have also deleted few words and sentences which convey unnecessary message and which were redundant.

Besides, we have complied the “Discussion” and “Conclusions” in only one section under the title “Discussion” and we have also thoroughly revised the concluding paragraph (conclusions) at the end of the discussion. Now we feel that the “Introduction and Discussion” are shortened with no more repetition of concepts.

Kind regards,

The authors.